Contents lists available at ScienceDirect

Heliyon

journal homepage: www.cell.com/heliyon

Digital financial inclusion and quality of economic growth

Wenzhi Xi^a, Yingdong Wang^{b,*}

^a School of Statistics and Mathematics, Zhongnan University of Economics and Law, Wuhan, 430073, China
 ^b School of Finance, Zhongnan University of Economics and Law, Wuhan, 430073, China

ARTICLE INFO

Keywords: Digital financial inclusion Quality of economic growth Entrepreneurial vitality Entropy weight method Spatial spillover effect

ABSTRACT

Recently, digital financial inclusion has experienced rapid growth, introducing transformative changes to the finance industry. Its attributes of low cost, extensive coverage, and efficient cross-temporal and spatial information dissemination have had a significant impact on both economic growth and social development. This study, using China's provincial panel data, measures the quality of economic growth based on technological innovation, coordinated development, environmental protection, opening to the outside world, and people's livelihood. It verifies the impact of digital financial inclusion on the quality of economic growth by utilizing the panel fixed effect model, mediation effect model, panel threshold model, and spatial Durbin model. Digital financial inclusion has a positive impact on the quality of economic growth, particularly in the eastern region and regions with high marketization. It can effectively promote the quality of economic growth by stimulating entrepreneurial vitality, and has had a positive and increasing non-linear effect on the quality of economic growth in neighboring regions. Therefore, the quality of economic growth in neighboring regions. Therefore, the quality of economic growth in neighboring regions.

1. Introduction

Recently, the Organization for Economic Cooperation and Development (OECD) indicated the untapped potential of the digital economy and the far-reaching implications of digital transformation for economic growth and social development. Especially, digital financial inclusion can promote innovation in the finance industry and meet the capital needs of the long-tail market, thereby improving the quality of economic growth [1–3]. Compared with traditional finance, digital financial inclusion emphasizes accurate risk control and effective information exchange using advanced tools and technologies such as the Internet, big data, and cloud computing, so that all social groups can enjoy equal, convenient, efficient, and affordable financial products and services [4,5]. This will produce historical changes in the finance industry and lead to greater economic development [6,7].

Digital financial inclusion encompasses a variety of financial services, such as mobile payment, online loan, online insurance, online fund, etc. On the one hand, it has expanded the coverage of financial services via the Internet and established a robust financial system, especially in underdeveloped regions with scarce financial resources [8]. On the other hand, it identifies customer information through big data, reduces the potential credit discrimination of traditional financial institutions against low-income groups, and improves the risk control ability of the finance industry [9]. Therefore, digital financial inclusion can provide more convenient and

https://doi.org/10.1016/j.heliyon.2023.e19731

Received 16 March 2023; Received in revised form 16 August 2023; Accepted 31 August 2023

Available online 5 September 2023







^{*} Corresponding author. School of Finance, Zhongnan University of Economics and Law, 182# Nanhu Avenue, Wuhan, 430073, Hubei, China. *E-mail addresses:* xiwenzhi1997@gmail.com (W. Xi), yingdongwang@126.com (Y. Wang).

^{2405-8440/© 2023} The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/).

effective financial services for all strata and groups of society. This is conducive to improving the efficiency of financial services for the real economy and sharing the fruits of economic development for all.

The advancement of digital financial inclusion holds immense potential for facilitating the provision of efficient, affordable, convenient, and accessible financial products and services to small and medium-sized enterprises and individuals with limited income. This advancement has the potential to improve the access to financial services in underdeveloped regions, while improving the efficacy of financial services for the real economy. Ultimately, these developments significantly foster high-quality economic development. Clarifying the mechanism of digital financial inclusion promoting the quality of economic growth is necessary to understand the vigorous development of digital financial inclusion, deepen the reform of the financial supply side, and accelerate economic transformation. It has considerable theoretical and practical significance for promoting digital financial inclusion and improving the quality of economic growth.

The objective of our study is to explore how digital financial inclusion affects the quality of economic growth? If the effect is confirmed, what is the underlying mechanism? What are the external conditions for the effect of digital financial inclusion on the quality of economic growth? What is the trend of the effect of digital financial inclusion on the quality of economic growth? Will the impact of digital financial inclusion on the quality of local economic growth have a radiation effect on the quality of economic growth in nearby areas? Although information technologies such as the Internet have made significant contributions to the society, and digital financial inclusion has gradually become a crucial part of the national economy, very little empirical research has been conducted to accurately assess the impact of digital financial inclusion on the quality of economic growth. Our study helps clarify the intrinsic mechanism and external conditions required for digital financial inclusion to affect the quality of economic growth. In this manner, this study provides valuable policy references regarding how to unleash the power of digital financial inclusion to drive the quality of economic growth. First, we identify the causal relationship and underlying mechanism between digital financial inclusion and the quality of economic growth. Second, we analyze the heterogeneity of digital financial inclusion on the quality of economic growth from the perspective of geographical location and marketization degree. Third, we discuss the non-linear and spatial spillover effects between them. In general, our research can further clarify the mechanisms of digital financial inclusion to promote the quality of economic growth. It has immense theoretical and practical significance for promoting the development of digital financial inclusion and improving the quality of economic growth.

Compared with the existing research, our study makes three significant contributions. First, previous studies have mainly discussed the impact of digital financial inclusion on financial efficiency [10], enterprise innovation [11], consumption structure [12,13], and sustainable development [14]. However, research on the impact of digital financial inclusion on the quality of economic growth is relatively scarce. We use the panel data of 30 provinces in China from 2011 to 2017 to explore the impact of digital financial inclusion on the quality of economic growth and its internal mechanism. We find that digital financial inclusion improves the quality of economic growth by stimulating entrepreneurial vitality, and this positive effect is more pronounced in the eastern as well as more market-oriented regions. Thus, our empirical evidence enriches the emerging literature on digital financial inclusion and the quality of economic growth.

Second, our study provides a feasible and comprehensive method for measuring the quality of economic growth. We construct an index system of economic growth quality from the five dimensions, and reasonably measure the quality of economic growth by combining the subjective weighting and the entropy weight method. Many previous studies predominantly focus on economic indicators when assessing the quality of economic growth, often overlooking crucial dimensions such as social, cultural, and environmental factors [15–17]. Our study aims to address this limitation by developing the index system that incorporates five dimensions, allowing for a more comprehensive and well-rounded measurement of the quality of economic growth. By incorporating these additional dimensions, our approach provides a more comprehensive and reasonable assessment of the multifaceted nature of economic growth quality.

Third, our research has significant policy implications for guiding the sustainable and well-regulated advancement of digital financial inclusion and fostering high-quality economic development. We find that digital financial inclusion has a significant and continuous positive impact on the quality of economic growth. Additionally, it has a positive effect on the quality of economic growth in neighboring regions. Therefore, government departments should guide the society to increase investment in information technology, particularly by expediting the commercialization of 5G networks, establishing comprehensive big data models, and leveraging artificial intelligence applications. These measures can effectively promote the development of digital financial inclusion, and thereby, contribute to more substantial endeavors toward high-quality economic development. Additionally, government departments should endeavor to coordinate the implementation of new infrastructure projects across various regions, thus maximizing the spatial spillover effect of digital financial inclusion on the quality of economic growth.

The rest of the article is organized as follows. Section 2 reviews the related literature and develops our main hypothesis. Section 3 illustrates the measurement of the quality of economic growth, and Section 4 discusses the data and models. Section 5 presents empirical results and discussions between digital financial inclusion and the quality of economic growth, while Section 6 reports the conclusions and implications.

2. Literature review

2.1. Digital financial inclusion

In recent years, the market scale of digital financial inclusion has continued to expand. Additionally, significant innovation has occurred regarding the types of products and services, showing a diversified, customized, and standardized financial development

pattern. Digital financial inclusion can be defined as achieving the goal of financial inclusion by vigorously developing digital finance to reduce the cost of financial services and expand the coverage of financial customers [18]. Das and Christie (2019) indicated that disruptive technology is the power source of digital financial inclusion [5]. Khera et al. (2022) found that the large-scale application of advanced technology in the finance industry and the popularization of digital financial services are the key driving forces of financial inclusion [19].

Many studies have investigated the inclusiveness, innovation, and risk control of digital financial inclusion. From the perspective of inclusiveness, Ren et al. (2018) argued that digital financial inclusion can effectively alleviate financial exclusion in society [20]. He and Li (2020) identified that online social interaction promotes the participation of rural households in digital financial inclusion, thus increasing the depth and breadth of digital financial inclusion [21]. From the perspective of innovation, Buchak et al. (2018) believed that financial technology (fintech) has filled the gaps in the traditional finance industry and energized financial markets [22]. Boot et al. (2021) suggested that the rapid popularization of digital financial inclusion may reconstruct the business model of traditional banks horizontally and vertically [7]. Li et al. (2023) opined that digital financial inclusion can address capital misallocation and improve financial efficiency, as well as promote green innovation of enterprises by mitigating the financing challenges associated with green projects [23]. Yang et al. (2022) confirmed that digital financial inclusion has a positive impact on enterprise innovation, creating opportunities for the implementation of enterprise innovation projects [18]. Hence, digital financial inclusion can induce historic changes to the finance industry and other industries by facilitating the adoption of novel technologies within the financial landscape [6].

From the perspective of risk control, Ozili (2018) indicated the prevalence of data security vulnerabilities in digital finance and the imperfection of the regulatory system of digital finance [4]. Yue et al. (2022) found that digital finance increases the risk of households falling into debt traps while achieving financial inclusion [24]. Yao et al. (2020) also confirmed that the fintech industry has high risks and is affected by both past developments and internal macroeconomic conditions [25]. However, Cheng and Qu (2020) employed the web crawler technology and word frequency analysis to construct and measure the bank fintech index, finding that bank fintech significantly reduces the credit risk of Chinese commercial banks [26]. Zhang et al. (2020) indicated that fintech can mitigate the pre-risk of credit activities and has no significant impact on post-loan risks related to credit activities [27]. Li et al. (2022) confirmed that bank fintech can improve operating income and capital adequacy ratio, optimize business performance, and improve risk control ability to reduce banks' risk-taking [28]. Generally, although digital financial inclusion compensates for the shortcomings of traditional finance, it also creates new financial risks such as transaction fraud, customer data leakage, and insufficient investor protection.

2.2. Quality of economic growth

Existing studies have mostly focused on the pace of economic growth based on the growth rate of the gross domestic product (GDP) [29,30], and discussed the impact of inflation [31], foreign trade [32], environmental changes [33–35], factor mobility [36], and happiness inequality [37] on economic growth. In the context of improved levels of economic development, deterioration of environmental problems, and widening of the rich-poor divide, the quality of economic growth has been receiving increasing academic attention.

In the study of the quality of economic growth, its measurement remains the primary issue. Recently, the relevant literature has also discussed the index system and the measurement method of the quality of economic growth. Huang et al. (2020) employed the improved G1 method based on the grey correlation degree to perform a five-dimensional measurement of the quality of economic growth: innovation development, urban-rural coordination, ecological environment, opening to the outside world, and people's livelihood [38]. Kong et al. (2021) measured the quality of economic growth based on the efficiency, stability, and sustainability of economic growth using the principal component analysis [39]. From the perspective of influencing factors, the existing research discusses the impact of industrial agglomeration [40], transportation infrastructure [41], environmental regulation [42–44], import technology [45], and other factors on the quality of economic growth.

2.3. The relationship between finance and economic growth

Sufficient research has been conducted on the finance-economic growth relationship, mainly including the financial deepening and financial repression theories. Agnello et al. (2012) found that financial reform is conducive to improving income inequality for economic development [46]. Asteriou and Spanos (2019) confirmed that pre-crisis financial development can promote economic growth, but post-crisis financial development can hinder economic activity [47]. Wong et al. (2021) indicated that the spatial agglomeration of financial services has a positive impact on the quality of urban economic growth [48]. Gao et al. (2022) identified that green finance reduces environmental pollution by adjusting the industrial structure and promoting technological progress, thereby improving the quality of economic growth [49]. Chen and He (2022) indicated that financial resilience is a necessary condition for promoting the quality of economic growth is a significant channel between financial resilience and the quality of economic growth [50].

However, research on digital financial inclusion and economic growth is scarce. Current research focuses on the influence of digital financial inclusion on consumption, income, and development imbalance. Liu et al. (2021) argued that digital financial inclusion makes a significant contribution to economic growth [51]. Li and Wu (2020) used the panel data of the China Household Finance Survey to conclude that digital financial inclusion can promote household consumption [12]. Yu et al. (2022) suggested that digital financial inclusion affects the consumption structure of rural residents by improving income, convenience, and liquidity [13]. Yu and Wang (2021) indicated that digital financial inclusion effectively reduces the regional development imbalance and the urban-rural

income gap [52]. Xu et al. (2023) confirmed that digital financial inclusion can improve the level of household income and improve risk resistance, which is immensely significant for reducing poverty and preventing its return [53].

2.4. Hypothesis development

Existing research widely acknowledges that information asymmetry and high transaction costs are significant obstacles impeding the ability of traditional finance to foster economic development [54–56]. From the perspective of the credit rationing theory, information asymmetry complicates the achievement of equilibrium between borrowers and lenders, thereby hindering traditional financial institutions from effectively supporting the growth of the real economy [57,58]. The transaction cost theory also highlights the notable deficiencies of traditional financial institutions in serving vulnerable groups, such as small and micro enterprises [59,60]. These deficiencies stem from the higher transaction costs and lower returns associated with providing services to these groups. By employing digital technology, digital financial inclusion presents a transformative shift from traditional finance, addressing the challenges posed by information asymmetry and transaction costs [61–65]. It fosters improved access and affordability of financial services, particularly benefiting vulnerable groups.

Based on the above theoretical analysis, digital financial inclusion affects the quality of economic growth from three aspects. First, the application of advanced technology in digital financial inclusion has resulted in the innovation of the products and services of traditional finance, reduced the cost of financial services, and expanded the coverage of financial customers, which is conducive to the sustainable development of the real economy [66]. Second, digital financial inclusion can integrate scattered resources in the society and meet customer needs in long-tail markets to effectively improve supply and demand imbalance and resource mismatch in financial markets. Additionally, it covers underdeveloped regions and low-income groups, balancing inter-regional economic development [52]. Third, digital financial inclusion can ease the liquidity constraints and introduce low-threshold investment products, tap potential consumer demand, and absorb idle funds in the society to promote economic development [13]. We propose the following research hypothesis:

H1. Digital financial inclusion is positively related to the quality of economic growth.

From the perspective of influence mechanism, numerous studies have identified that digital financial inclusion provides funding sources for entrepreneurial activities and stimulates diversified demand in the market, which promotes new enterprises [51,67,68]. Furthermore, the enhancement of entrepreneurship also helps drive the quality of economic growth. Entrepreneurship is an endogenous force for economic growth and a significant driver of employment, industrial restructuring, environmental protection, etc. [69–71]. Therefore, digital financial inclusion promotes the quality of economic growth by stimulating entrepreneurial vitality in the society [72]. The second research hypothesis is as follows:

H2. Digital financial inclusion stimulates entrepreneurial vitality in the society, thereby promoting the quality of economic growth.

Tobler's First Law of Geography states that all things are spatially correlated with other things. Many studies have confirmed that digital financial inclusion has the characteristics of cross-space-time transmission and significant spatial effects [73–75]. Owing to the continuous expansion of the Internet and big data, digital financial inclusion has spillover effects on other industries, resulting in greater value for participants. Thus, it may have a non-linear or even geometric impact on the quality of economic growth. Additionally, as stated above, digital financial inclusion has the characteristics of cross-space-time transmission. It has significantly improved the efficiency of information exchange and expanded the coverage of financial services through modern information technology. This has strengthened the spatial correlation of economic activities between regions and promoted the quality of economic

Table 1

High-Quality Development Index evaluation system.

Primary indicators	Secondary indicators	Tertiary indicators
Innovation (Technological innovation)	Innovation input	R&D funding input intensity
-	-	R&D personnel input intensity
	Innovation output	Number of patents granted per 10,000 people
		Proportion of income of high-tech industry
Coordination (Coordinated development)	Industry coordination	Industrial structure rationalization index
		Industrial structure advanced index
	Urban-rural coordination	Proportion of consumption level of urban and rural residents
		Proportion of urban and rural income level
Green (Environmental protection)	Green life	Forest coverage rate
		Harmless treatment rate of household garbage
	Energy consumption	Energy consumption per unit of GDP
		Electricity consumption per unit of GDP
	Environmental governance	Comprehensive utilization rate of general industrial solid waste
		Investment in environmental pollution control as a proportion of GDP
Open (Opening to the outside world)	Foreign trade	Total imports and exports as a percentage of GDP
	Foreign investment	Foreign direct investment as a percentage of GDP
Sharing (People's livelihood)	Economic sharing	GDP per capita
		Per capita expenditure on public services
	Social sharing	Number of health institutions
		Per capita education expenditure

growth. Consequently, digital financial inclusion has a positive spatial spillover effect on the quality of economic growth. We propose Hypothesis 3, as follows:

H3. Digital financial inclusion has a positive and increasing impact on the quality of economic growth, and a positive spatial spillover effect on the quality of economic growth in neighboring regions.

3. Measurement of the quality of economic growth in China

The quality of economic growth implies not only GDP growth, but also technological innovation, coordinated development, environmental protection, opening to the outside world, and people's livelihood. Drawing on the ideas of Huang et al. (2020) and Lin and Zhou (2022) [38,76], this study constructs the High-Quality Development Index based on the five dimensions of innovation, coordination, green, open, and sharing, to evaluate the quality of economic growth in China. The index evaluation system includes five primary indicators, 11 secondary indicators, and 20 tertiary indicators (Table 1), encompassing the economy, the environment, life, etc.

The subjective and objective methods are applied to measure the quality of economic growth. First, the entropy weight method is used to assign weight to each tertiary indicator. It serves to calculate the information entropy value of each evaluation index based on the original observation value, and measure the relative change degree of the index with the information entropy value to determine the objective weight of the index [77]. Second, the five primary indicators are assigned 20% subjective weight, which shows that the five aspects have the same importance. Finally, the specific index score of each evaluation object is multiplied by the subjective weight and objective weight of the index, and the final score of the evaluation object is obtained. This study uses China's provincial panel data from 2011 to 2017 to measure the high-quality development index of China's provinces, including the total index and the indexes of five dimensions. All data in Table 1 have been derived from the Easy Professional Superior (EPS) database and the China Statistical Yearbook.

4. Data and methodology

4.1. Data

China's provincial panel data from 2011 to 2017 are adopted in this study. Excluding Tibet, where there is a lack of data, 30 provinces (cities, autonomous regions) have been selected as research samples, and 210 sample values have been obtained. The data used in this study are obtained mainly from the EPS database, China Statistical Yearbook, China Basic Statistical Units Yearbook, Statistical Reports on Internet Development in China, and Peking University Digital Inclusive Finance Index of China.¹ The Peking University Digital Financial Inclusion Index of China has been jointly compiled by the Institute of Digital Finance of Peking University and the Ant Financial Services Group.

4.2. Model specification

To prevent endogeneity problems caused by omitted variables, this study uses the panel fixed effect model² (Eq. (1)) to analyze the impact of digital financial inclusion on the quality of economic growth. We also included the province fixed effects and year fixed effects to capture regional-specific and year-specific factors.

$$HQD_{i,t} = \beta_0 + \beta_1 DFI_{i,t} + \beta_2 X_{i,t} + \delta_i + \delta_i + \varepsilon_{i,t}$$

$$\tag{1}$$

where, the dependent variable $HQD_{i,t}$ is the quality of economic growth of province *i* in year *i*. Additionally, innovation $(INN_{i,t})$, coordination $(COO_{i,t})$, green $(GRE_{i,t})$, open $(OPE_{i,t})$, and sharing $(SHA_{i,t})$ are also taken as dependent variables for the regression analysis. The independent variable $DFI_{i,t}$ signifies digital financial inclusion index. $DFI_BRE_{i,t}$ and $DFI_DEP_{i,t}$ are also used as independent variables to analyze the effects of the coverage breadth and usage depth of digital financial inclusion on the quality of economic growth. $X_{i,t}$ represents the set of control variables, δ_i controls the province fixed effects, δ_t controls the year fixed effects, and $\varepsilon_{i,t}$ is the clustering robust standard error.

To further test the influence mechanism between them, the following mediating effect model (Eq. (2) and (3)) is constructed:

$$ENT_{i,t} = \beta_0 + \beta_1 DFI_{i,t} + \beta_2 X_{i,t} + \delta_i + \delta_t + \varepsilon_{i,t}$$
⁽²⁾

$$HQD_{i,t} = \beta_0 + \beta_1 DFI_{i,t} + \beta_2 ENT_{i,t} + \beta_3 X_{i,t} + \delta_i + \delta_t + \varepsilon_{i,t}$$
(3)

where, $ENT_{i,t}$ is entrepreneurial vitality, measured by the ratio of the number of corporate units established each year to the local workforce aged 15 to 64. Considering that there may be a non-linear effect between digital financial inclusion and the quality of

¹ Specifically, the number of corporate units established each year is derived from the China Basic Statistical Units Yearbook. The Internet penetration rate is derived from the Statistical Reports on Internet Development in China. The data for the High–Quality Development Index and other variables are derived from the EPS database and the China Statistical Yearbook

² We also conducted the Hausman test and obtained a p value of 0.0000. So, the original hypothesis is strongly rejected and the fixed effect model should be adopted.

economic growth, this study builds the following single threshold model by referring to the panel threshold model (Eq. (4)) proposed by Hansen (1999):

$$HQD_{i,t} = \beta_0 + \beta_1 DFI_{i,t} I \left(DFI_{i,t} \le \gamma \right) + \beta_2 DFI_{i,t} I \left(DFI_{i,t} > \gamma \right) + \beta_3 X_{i,t} + \delta_i + \delta_t + \varepsilon_{i,t}$$

$$\tag{4}$$

where, $I(\cdot)$ is the indicator function that takes the value of 1 if the conditions in parentheses are met, and 0 otherwise. $DFI_{i,t}$ is the threshold variable and γ is the threshold value to be estimated. Hansen's (1999) model requires that the independent variable be strongly exogenous to ensure the consistency of the estimators [78]. Therefore, we use the first-differenced GMM (FD-GMM) estimation to address the possible endogeneity of the panel threshold model [79,80].

To further analyze the spatial spillover effect of digital financial inclusion on the quality of economic growth, we construct the spatial Durbin model (Eq. (5)).

$$HQD_{i,t} = \beta_0 + \rho W HQD_{i,t} + \varphi_1 W D F I_{i,t} + \varphi_2 W X_{i,t} + \beta_1 D F I_{i,t} + \beta_2 X_{i,t} + \delta_i + \delta_i + \delta_i + \delta_i + \delta_i$$

$$\tag{5}$$

where, ρ is the spatial autoregressive coefficient of the quality of economic growth, *W* is the spatial weight matrix, φ_1 is the coefficient of the spatial interaction term of digital financial inclusion, $X_{i,t}$ is a series of control variables, and φ_2 is the coefficient of the spatial interaction term of a series of control variables. Additionally, we select the geographical weight matrix to construct the spatial Durbin model.

4.3. Variable selection and definition

In our study, the dependent variable is the quality of economic growth in each region, which is measured by the High-Quality Development Index calculated in Section 3. The independent variable is the degree of development of digital financial inclusion, measured using the Digital Financial Inclusion Index. The index includes the Digital Financial Inclusion Index, Digital Financial Coverage Breadth Index, and Digital Financial Usage Depth Index. Among them, the coverage breadth of digital financial inclusion is measured based on the account coverage of digital finance. The usage depth of digital finance is measured based on the specific use of digital financial services by users, including payment, credit, monetary fund, insurance, credit, and investment business. Additionally, we select human capital, government intervention level, infrastructure level, and capital investment ratio as control variables. All variables are defined in Table 2.

4.4. Descriptive statistics

Table 3 shows descriptive statistics for all variables. During the sample period, the average value of the *HQD* index and its five dimensions of 30 provinces in China are 4.740, 1.193, 0.591, 1.430, 0.522, and 1.004 respectively, and the difference between their maximum and minimum values is significant. The mean values of the total index, coverage breadth, and usage depth of digital financial inclusion are 172.059, 151.497, and 168.576, respectively. The average values of human capital, government intervention level, infrastructure level, and capital investment ratio are 9.228, 0.245, 0.921, and 0.793 respectively.

4.5. Correlation analysis

To analyze the correlation between variables, we use the Pearson correlation test. As shown in Table 4, there is a correlation between the dependent variable and the independent variable. Additionally, the correlation coefficients between variables are not more than 0.8, so there is no high correlation between variables. This can well avoid multicollinearity in the regression models.

Table 1	2
---------	---

Variable definitions.				
Variable types	Variables	Symbols	Variable explanations	Units
Dependent variables	Quality of economic growth	HQD	High-Quality Development Index	-
	Innovation	INN	Innovation Index	-
	Coordination	COO	Coordination Index	-
	Green	GRE	Green Index	-
	Open	OPE	Open Index	-
	Sharing	SHA	Sharing Index	-
Independent variables	Digital financial inclusion	DFI	Digital Financial Inclusion Index	-
	Digital financial coverage breadth	DFI_BRE	Digital Financial Coverage Breadth Index	-
	Digital financial usage depth	DFI_DEP	Digital Financial Usage Depth Index	-
Mediator variable	Entrepreneurial vitality	ENT	Number of corporate units established/local workforce	-
Control variables	Human capital	HUM	Per capita education years	Year
	Government intervention level	GOV	Fiscal expenditure/GDP	%
	Infrastructure level	INF	Road mileage/land area	1/10,000 km
	Capital investment ratio	CAP	Capital investment/GDP	%

Table 3

Descriptive statistics.

1					
Variables	Observations	Mean	Std.Dev.	Min	Max
HQD	210	4.740	2.188	1.944	13.915
INN	210	1.193	1.107	0.118	6.301
COO	210	0.591	0.186	0.257	1.159
GRE	210	1.430	0.298	0.675	2.094
OPE	210	0.522	0.595	0.055	2.760
SHA	210	1.004	0.419	0.351	2.901
DFI	210	172.059	77.742	18.330	336.651
DFI_BRE	210	151.497	74.543	1.960	316.118
DFI_DEP	210	168.576	78.517	6.760	396.049
ENT	210	17.743	11.742	3.341	61.778
HUM	210	9.228	0.927	7.589	13.227
GOV	210	0.245	0.102	0.110	0.627
INF	210	0.921	0.494	0.089	2.101
CAP	210	0.793	0.243	0.240	1.469

Note: 1. *HQD* is the quality of economic growth, in which *INN*, *COO*, *GRE*, *OPE*, and *SHA* are innovation, coordination, green, open, and sharing, respectively. 2. *DFI*, *DFI_BRE*, and *DFI_DEP* are digital financial inclusion, digital financial coverage breadth, and digital financial usage depth, respectively. 3. *Hum* is human capital, *Gov* is government intervention level, Inf is infrastructure level, and *Cap* is capital investment ratio.

5. Results and discussion

5.1. Baseline regression results

Table 5 shows the baseline regression results of the panel fixed effect model. From Panel A, the regression coefficients of digital financial inclusion on the quality of economic growth and its five dimensions are all positive at the 1% significant level. Therefore, digital financial inclusion not only has a significant positive impact on the quality of economic growth, but also promotes technological innovation, coordinated development, environmental protection, opening to the outside world, and people's livelihood. Panels B and C present the regression results with *DFI_BRE* and *DFI_DEP* as explanatory variables, respectively. The regression results present that the improvement of coverage breadth and usage depth of digital financial inclusion can significantly promote the quality of economic growth and the development of its five areas. There may be two reasons for the results. First, the coverage breadth of digital financial inclusion refers to the account coverage of digital finance. The improvement of the coverage breadth can better alleviate financial exclusion and meet the financial needs of underdeveloped areas and low-income people, so as to effectively address the problem of uncoordinated and unbalanced development. Second, the usage depth of digital financial inclusion is reflected in the use of digital financial services such as payments, credit, money funds, insurance, credit, and investment business. The continuous increase of the usage depth can further establish a perfect digital financial inclusive system and better support the development of the real economy. The above results suggest that vigorously developing digital financial inclusion is an effective method of promoting the quality of economic growth.

To address the possible endogenous problems, the number of fixed telephone users (*IV1*), per capita post and telecommunications business volume (*IV2*), and Internet penetration rate (*IV3*) are selected as instrumental variables for the two-stage least squares (2SLS) regression. On the one hand, they have a strong correlation with digital financial inclusion. On the other hand, they have little direct impact on the quality of economic growth. Therefore, they are effective instrumental variables. When using the above three instrumental variables for the 2SLS regression, digital financial inclusion has a significant positive correlation with them. As shown in Table 6, the regression coefficients of digital financial inclusion are positive at the 1% significant level. After addressing the endogenous problems, our results remain robust.

Table 4	
Correlation coefficient test.	

	HQD	DFI	ENT	HUM	GOV	INF	CAP
HQD	1.000						
DFI	0.434***	1.000					
ENT	0.672***	0.753***	1.000				
HUM	0.782***	0.418***	0.484***	1.000			
GOV	-0.415***	-0.064	-0.146**	-0.368***	1.000		
INF	0.604***	0.260***	0.386***	0.426***	-0.588***	1.000	
CAP	-0.681^{***}	0.079	-0.151**	-0.581***	0.564***	-0.477***	1.000

Note: 1. *HQD* is the quality of economic growth. 2. *DFI* is digital financial inclusion. 3. *Hum* is human capital, *Gov* is government intervention level, Inf is infrastructure level, and *Cap* is capital investment ratio. 4. ***, **, and * denote significance at the 1%, 5%, and 10% levels, respectively, with standard errors in parentheses.

Table 5

Baseline regression results.

Variables	HOD (1)	INN (2)	COO (3)		GRE (4)	OPE (5)		SHA (6)	
Panel A: Regress	sion results of DFI as	s the explanatory va	riable					- (-)	
DFI	0.068***	0.026***	0.003***	r	0.010***		0.017***	0.012**	k
	(0.0048)	(0.0034)	(0.0004)		(0.0011)		(0.0015)	(0.0010)
HUM	0.477***	0.175**	0.089***	r -	-0.085***		0.076**	0.221**	k
	(0.1020)	(0.0845)	(0.0101)		(0.0247)		(0.0353)	(0.0212)	1
GOV	3.542***	0.836	0.387***	e e e e e e e e e e e e e e e e e e e	-1.526***		1.924***	1.922**	*
	(0.7795)	(0.6036)	(0.0638)		(0.2221)		(0.2129)	(0.1964)	1
INF	0.476***	0.408***	0.024*		-0.138***		0.205***	-0.023	,
	(0.1227)	(0.0946)	(0.0138)		(0.0401)		(0.0427)	(0.0396)	1
CAP	-1.276**	-0.977***	-0.128*	**	0.379***		-0.760***	0.210*	
	(0.5320)	(0.3394)	(0.0427)		(0.0963)		(0.1433)	(0.1073)	1
Constant	-3.275***	-1.497*	-0.469*	**	1.917***		-0.919**	-2.307*	**
constant	(1,1374)	(0.8947)	(0.1085)		(0.2623)		(0.3704)	(0.2204)	1
R^2	0.902	0.779	0.850		0.457		0.859	0.825	,
Obs	210	210	210		210		210	210	
Panel B. Regress	ion results of DFI B	RF as the explanato	rv variahl	۵	210		210	210	
DFI	0.068***	0.026***	0.003***		0.010***		0.017***	0.012**	k
DII	(0.0048)	(0.0034)	(0.0004)		(0.0011)		(0.0015)	(0.0010	1
HUM	0 477***	0 175**	0.089***		-0.085***		0.076**	0 221**	k
nom	(0, 1020)	(0.0845)	(0.0101)		(0.0247)		(0.0353)	(0.0212)	1
GOV	3.542***	0.836	0.387***		-1.526***		1.924***	1 922**	, k
007	(0.7795)	(0.6036)	(0.0638)		(0.2221)		(0.2129)	(0.1964)	1
INF	0.476***	0 408***	0.024*		-0.138***		0.205***	-0.023	,
	(0.1227)	(0.0946)	(0.0138)		(0.0401)		(0.0427)	(0.0396)	1
CAP	-1.276**	-0.977***	-0.128*	**	0.379***		-0.760***	0.210*	,
Ci li	(0.5320)	(0.3394)	(0.0427)		(0.0963)		(0.1433)	(0.1073)	1
Constant	-3.275***	-1.497*	-0.469*	**	1.917***		-0.919**	-2.307*	**
	(1.1374)	(0.8947)	(0.1085)		(0.2623)		(0.3704)	(0.2204)	1
R^2	0.902	0.779	0.850		0.457		0.859	0.825	,
Obs	210	210	210		210		210	210	
Panel C: Regress	ion results of DFI D	EP as the explanato	rv variabl	e	210		210	210	
DFI DEP	0.033***	0.012***	19 1011001	0.002***	0.006***		0.008***		0.005***
211_2221	(0.0036)	(0.0022)		(0.0002)	(0.0007)		(0,0009)		(0,0009)
HUM	0.896***	0.334***		0.106***	-0.025		0.183***		0.297***
	(0.1303)	(0.0919)		(0.0111)	(0.0263)		(0.0379)		(0.0232)
GOV	3.917***	0.980		0.415***	-1.431***		1.999***		1.954***
	(0.8736)	(0.6217)		(0.0669)	(0.2355)		(0.2341)		(0.2229)
INF	0.466***	0 402***		0.015	-0.169***		0.217***		0.001
	(0.1408)	(0.0874)		(0.0142)	(0.0439)		(0.0537)		(0.0445)
CAP	-1.996***	-1.243***		-0.135***	0.350***		-0.981***		0.014
	(0.6791)	(0.3915)		(0.0387)	(0.1030)		(0.1827)		(0.1396)
Constant	-5.405***	-2.306**		-0.583***	1.529***		-1.424***		-2.621***
	(1.5240)	(1.0148)		(0.1152)	(0.2928)		(0.4339)		(0.2775)
R^2	0.855	0.754		0.851	0.459		0.808		0.761
Obs	210	210		210	210		210		210
				-			-		

Note: 1. *HQD* is the quality of economic growth, in which *INN*, *COO*, *GRE*, *OPE*, and *SHA* are innovation, coordination, green, open, and sharing, respectively. 2. *DFI*, *DFI_BRE*, and *DFI_DEP* are digital financial inclusion, digital financial coverage breadth, and digital financial usage depth, respectively. 3. *Hum* is human capital, *Gov* is government intervention level, Inf is infrastructure level, and *Cap* is capital investment ratio. 4.***, **, and * denote significance at the 1%, 5%, and 10% levels, respectively, with standard errors in parentheses.

5.2. Heterogeneity analysis

Owing to differences in geographic location and degree of marketization, digital financial inclusion of each province presents different development patterns. The 30 provinces are grouped and regressed separately according to the differences in geographical location and marketization index (see Table 7). From Columns (1) to (3), compared with the central and western regions, the positive effect of digital financial inclusion on the quality of economic growth is greater in the eastern region. In Columns (4) and (5), provinces with higher marketization have a greater role in promoting the quality of economic growth than those with lower marketization. As provinces with higher marketization have better market mechanisms, digital financial inclusion can better optimize the allocation of resources in these markets.

5.3. Mechanism analysis

This study uses Eqs. (2) and (3) to test the influence mechanism (see Table 8). In Column (1), digital financial inclusion has a significant positive impact on entrepreneurial vitality. In Column (2), the coefficients of digital financial inclusion and entrepreneurial vitality are 0.050 and 0.076, respectively, at the 1% significance level. This shows that digital financial inclusion promotes the quality

Table 6		
Instrumental	variables	regression.

	IV1	IV2	IV3
Variables	HQD (1)	HQD (2)	HQD (3)
DFI	0.219***	0.098***	0.081***
	(0.0535)	(0.0071)	(0.0068)
HUM	-0.635	0.254**	0.381***
	(0.4257)	(0.1107)	(0.1207)
GOV	4.965**	3.827***	3.664***
	(1.9716)	(0.8863)	(0.7859)
INF	-1.200*	0.141	0.333**
	(0.6822)	(0.1725)	(0.1579)
CAP	4.863*	-0.047	-0.751
	(2.5413)	(0.4000)	(0.4629)
Constant	-2.392	-3.099***	-3.200***
	(2.3501)	(1.0758)	(1.0801)
R^2	0.263	0.877	0.897
Obs	210	210	210

Note: 1. *IV1*, *IV2*, and *IV3* are the number of fixed telephone users, per capita post and telecommunications business volume, and Internet penetration rate, respectively. 2. *HQD* is the quality of economic growth. 3. *DFI* is digital financial inclusion. 4. *Hum* is human capital, *Gov* is government intervention level, Inf is infrastructure level, and *Cap* is capital investment ratio. 5.***, **, and * denote significance at the 1%, 5%, and 10% levels, respectively, with standard errors in parentheses.

Table 7

Heterogeneity analysis.

	East	Central	West	Higher marketization	Lower marketization
Variables	HQD (1)	HQD (2)	HQD (3)	HQD (4)	HQD (5)
DFI	0.064***	0.035***	0.027***	0.062***	0.040***
	(0.0091)	(0.0054)	(0.0053)	(0.0076)	(0.0044)
HUM	0.804***	-0.340***	0.346***	0.701***	0.183***
	(0.1599)	(0.1255)	(0.0494)	(0.1520)	(0.0620)
GOV	-4.166**	-6.675***	0.309	-1.658	0.535
	(1.8622)	(2.0686)	(0.5617)	(2.2165)	(0.5158)
INF	0.059	-0.545***	0.669***	0.143	0.195
	(0.4085)	(0.1895)	(0.0702)	(0.3073)	(0.1179)
CAP	-1.560	0.865***	0.249	-1.509	0.316
	(1.2237)	(0.3088)	(0.3182)	(1.0078)	(0.2918)
Constant	-3.949**	6.447***	-1.460***	-3.347*	-0.326
	(1.9461)	(1.3410)	(0.4811)	(1.8347)	(0.6239)
R^2	0.885	0.763	0.912	0.892	0.792
Obs	77	56	77	105	105

Note: 1. *HQD* is the quality of economic growth. 2. *DFI* is digital financial inclusion. 3. *Hum* is human capital, *Gov* is government intervention level, Inf is infrastructure level, and *Cap* is capital investment ratio. 4.***, **, and * denote significance at the 1%, 5%, and 10% levels, respectively, with standard errors in parentheses.

of economic growth by stimulating local entrepreneurial vitality.

5.4. Non-linear effect analysis

The threshold variable *DFI* passes a single threshold test using the bootstrap method with 1000 replications, and the threshold value is 285.277. There is a non-monotonic relationship between digital financial inclusion and the quality of economic growth. Then, *DFI_BRE* and *DFI_DEP* replace *DFI*, and both tests pass significantly. The results of the panel threshold model in Table 9 show that the coefficient of digital financial inclusion is 0.004 points below the threshold (*DFI* \leq 285.277), and the one is 0.006 above the threshold value (*DFI* >285.277). It can be seen that the higher the degree of development of digital financial inclusion, the greater the promotion effect on the quality of economic growth. In Columns (2) and (3), its coverage breadth and usage depth also show a positive and increasing non-linear effect on the quality of economic growth over the past few years.

The panel model with threshold effects proposed by Hansen (1999) has been widely used in empirical research [78], but the fixed effect estimator of the model requires that the covariates are strongly exogenous for the estimator to be consistent. However, this strong externality may be restrictive; therefore, Seo and Shin (2016) proposed the FD-GMM estimation for addressing endogenous independent variables and endogenous threshold variables [80]. In Table 10, the number of fixed telephone users is treated as a tool variable to address the endogeneity problem using the FD-GMM estimation. The results show that the regression coefficient of digital financial inclusion above the threshold value is higher than its coefficient below the threshold value. This further confirms the

Variables	ENT (1)	HQD (2)
DFI	0.231***	0.050***
	(0.0358)	(0.0038)
ENT		0.076***
		(0.0058)
HUM	0.355	0.450***
	(0.9726)	(0.0683)
GOV	11.338	2.683***
	(7.9529)	(0.4215)
INF	2.733**	0.269***
	(1.2261)	(0.0892)
CAP	1.010	-1.352^{***}
	(4.5138)	(0.3324)
Constant	-9.964	-2.520***
	(10.6475)	(0.6917)
R^2	0.712	0.950
Obs	210	210

Note: 1. *ENT* is entrepreneurial vitality, and *HQD* is the quality of economic growth. 2. *DFI* is digital financial inclusion. 3. *Hum* is human capital, *Gov* is government intervention level, Inf is infrastructure level, and *Cap* is capital investment ratio. 4.***, **, and * denote significance at the 1%, 5%, and 10% levels, respectively, with standard errors in parentheses.

Table 9

Estimation results of the panel threshold model.

Table 8

Variables	DFI (1)	DFI_BRE (2)	DFI_DEP (3)
Threshold value γ	285.277***	268.390***	317.581***
$DFI*I (DFI \leq \gamma)$	0.004*** (0.0005)		
$DFI*I (DFI > \gamma)$	0.006*** (0.0005)		
DFI_BRE*I ($DFI_BRE \leq \gamma$)		0.004*** (0.0005)	
DFI_BRE*I ($DFI_BRE > \gamma$)		0.007*** (0.0005)	
DFI_DEP*I ($DFI_DEP \leq \gamma$)			0.003*** (0.0004)
DFI_DEP*I ($DFI_DEP > \gamma$)			0.004*** (0.0004)
Control variables	YES	YES	YES
R^2	0.826	0.848	0.797
Obs	210	210	210

Note: 1. *HQD* is the quality of economic growth. 2. *DFI*, *DFI*_*BRE*, and *DFI*_*DEP* are digital financial inclusion, digital financial coverage breadth, and digital financial usage depth, respectively. 3. *I* (\cdot) is the indicator function that takes the value of 1 if the conditions in parentheses are met, and 0 otherwise. 4. γ is the threshold value to be estimated. 5.Control variables include human capital, government intervention level, infrastructure level, and capital investment ratio. 6.***, **, and * denote significance at the 1%, 5%, and 10% levels, respectively, with standard errors in parentheses.

Table 10

FD-GMM estimation results.

Variables	DFI (1)	DFI_BRE (2)	DFI_DEP (3)
$DFI*I$ ($DFI < \gamma$)	0.001 (0.0007)		
$DFI^{T}(DFI > \gamma)$	0.003^^ (0.0012)		
DFI_BRE*I ($DFI_BRE \leq \gamma$)		-0.003** (0.0014)	
DFI_BRE*I ($DFI_BRE > \gamma$)		0.006*** (0.0018)	
DFI_DEP*I ($DFI_DEP \leq \gamma$)			0.001*** (0.0003)
DFI_DEP*I ($DFI_DEP > \gamma$)			0.004*** (0.0015)
Control variables	YES	YES	YES
Number of moment conditions	51	51	51
Obs	210	210	210

Note: 1. *HQD* is the quality of economic growth. 2. *DFI*, *DFI_BRE*, and *DFI_DEP* are digital financial inclusion, digital financial coverage breadth, and digital financial usage depth, respectively. 3. $I(\cdot)$ is the indicator function that takes the value of 1 if the conditions in parentheses are met, and 0 otherwise. 4. γ is the threshold value to be estimated. 5.Control variables include human capital, government intervention level, infrastructure level, and capital investment ratio. 6.***, **, and * denote significance at the 1%, 5%, and 10% levels, respectively, with standard errors in parentheses.

non-linear characteristics of the increasing marginal effect of digital financial inclusion.

5.5. Spatial econometric analysis

To analyze the spatial spillover effect of digital financial inclusion on the quality of economic growth, we construct the spatial Durbin model. Before using spatial econometric methods, it is necessary to test the spatial autocorrelation of the quality of economic growth. We use the geographical weight matrix to calculate *Moran's I* of the annual quality of economic growth, and find that they are significantly greater than 0. This shows a spatial positive correlation between the quality of economic growth in China, that is, the regions with higher quality of economic growth come together.

Considering the limited significance of the regression coefficients of the explanatory variables of the spatial econometric model, we further explain the direct and indirect effects of the spatial Durbin model. As shown in Table 11, digital financial inclusion not only has a significant positive direct effect on the quality of economic growth, but also has a significant positive spatial spillover effect (indirect effects) on the quality of economic growth in neighboring regions. The possible reason is that digital financial inclusion has the characteristics of cross-temporal and spatial information dissemination and significant network effects. Therefore, the development of local digital financial inclusion can promote the quality of economic growth in neighboring regions.

5.6. Discussion and research implications

Recent studies have mainly explored the positive role of digital financial inclusion in reducing regional development imbalance, narrowing the urban-rural income gap, protecting the environment, and promoting residents' consumption [12,13,81–83]. First, different from the existing research, we confirm a positive correlation between digital financial inclusion and the quality of economic growth. Additionally, digital financial inclusion promotes technological innovation, coordinated development, environmental protection, opening to the outside world, and people's livelihood. This indicates that digital financial inclusion is a crucial factor in the promotion of the quality of economic growth. Numerous studies have identified that digital financial inclusion is conducive to promoting economic growth, reducing carbon dioxide emissions, alleviating poverty, etc. [51,75,84]. However, a few studies have indicated that although digital financial inclusion can stimulate economic growth, it can also cause a surge in carbon dioxide emissions and reduce environmental quality [85]. Our research supports the argument that digital financial inclusion can achieve a win-win situation between economic development and environmental protection, and comprehensively improve the quality of economic growth. Therefore, governmental departments must encourage the application of advanced science and technology in the financial field, and provide sufficient trial and error opportunities and a good innovation environment for the development of digital financial inclusion.

Second, stimulating entrepreneurial vitality is significant for digital financial inclusion to promote the quality of economic growth. Additionally, the positive effect of digital financial inclusion on the quality of economic growth is more significant in the eastern region and regions with high degree of marketization. This is probably because a robust information infrastructure and effective market mechanisms can better deliver the advantages of digital financial inclusion, resulting in greater economic development. Xu et al. (2023) found that digital financial inclusion improves household income levels by promoting entrepreneurship and employment [53]. Yang et al. (2022) suggested that digital financial inclusion improves women's job flexibility and encourages their entrepreneurship [67]. According to Chen et al. (2022), the development of digital financial inclusion requires a solid foundation of the traditional financial system to fully play its role [86]. These studies support our results to some extent. Therefore, governmental departments must accelerate the construction of information infrastructure and foster a favorable market environment.

Finally, digital financial inclusion can promote the quality of economic growth in neighboring regions. Many studies have confirmed that digital financial inclusion and the quality of economic growth have significant spatial autocorrelation [74,87], and digital financial inclusion has a positive spatial spillover effect on farmers' income growth [88]. Aziz and Naima (2021) concluded that although digital services have improved the access to financial services, these services have not been fully utilized due to a lack of basic connectivity, financial literacy, and social awareness [89]. This implies that government and financial institutions must strengthen the exchange and cooperation of digital financial inclusion between regions. Additionally, they must improve the efficiency of resource allocation between regions to exert the spatial spillover effect of digital financial inclusion on the quality of economic growth.

Our study provides substantial empirical evidence to show that digital financial inclusion promotes the quality of economic growth. However, there are two limitations. First, we use the Peking University Digital Financial Inclusion Index of China to measure digital financial inclusion. However, owing to the unavailability of data and the unmatched data of different institutions, the index only uses the data of a representative digital financial institution. Therefore, it cannot fully reflect the integrity of the development of digital financial inclusion. Second, we find that digital financial inclusion can promote the quality of economic growth by stimulating entrepreneurial vitality. However, there may be more potential transmission mechanisms between digital financial inclusion and the quality of economic growth, which need to be further explored.

In general, our study has significant research implications. First, we establish a systematic, scientific, typical, dynamic, and operable index evaluation system for the quality of economic growth. Additionally, we employ the more objective entropy method to measure the quality of economic growth in China's provinces, and thereafter, provide a feasible measure for the empirical study of the quality of economic growth. Second, we explore the transmission mechanism and spatial spillover effect of digital financial inclusion on the quality of economic growth, which is conducive to enriching the theoretical discussion and empirical evidence in this field. Third, our study is helpful to understand the historical opportunity of the rapid development of science and technology, vigorously promote digital financial inclusion to improve the efficiency of financial institutions and financial markets, promote the

Variables	Direct effect	Indirect effect	Total effect
	HQD (1)	HQD (2)	HQD (3)
DFI	0.002*	0.002**	0.003***
	(0.0010)	(0.0007)	(0.0005)

 Table 11

 Spatial effect estimation results.

Note: 1. *HQD* is the quality of economic growth. 2. *DFI* is digital financial inclusion. 3. ***, **, and * denote significance at the 1%, 5%, and 10% levels, respectively, with standard errors in parentheses.

transformation and upgrading of the finance industry, and make digital financial inclusion better promote economic and social development.

6. Conclusions and policy recommendations

6.1. Conclusions

Based on the provincial panel data of China from 2011 to 2017, this study combines the subjective weighting method and the entropy weight method to measure the quality of economic growth. Additionally, it empirically tests the impact of digital financial inclusion on the quality of economic growth by utilizing the mediation effect model, the panel threshold model based on the FD-GMM estimation, and the spatial Durbin model. The conclusions are as follows. First, digital financial inclusion has a significant positive impact on the quality of economic growth, which is more obvious in the eastern region and regions with high marketization. Second, digital financial inclusion can stimulate entrepreneurial vitality, thereby improving the quality of economic growth. Third, digital financial inclusion and its coverage breadth and usage depth have had a positive and increasing non-linear effect on the quality of economic growth, but also promote the quality of economic growth in neighboring regions. Therefore, the quality of economic development can be significantly improved by continuously developing digital financial inclusion.

6.2. Policy recommendations

Based on our findings and the actual situation of digital financial inclusion, we propose policy recommendations to promote digital financial inclusion and the quality of economic growth. These can be used by the government to formulate policies aligning with local advantages and guide the finance industry to adopt suitable development strategies.

First, it is essential to encourage digital financial inclusion innovation and vigorously develop related technologies and business models of digital financial inclusion. The technological and business model innovations are conducive to meeting the financial needs of small enterprises and low-income groups, thereby expanding the customer base and market share of digital financial inclusion. Therefore, government departments must encourage the attempt and application of advanced science and technology in the financial field, and provide sufficient trial and error space and a favorable innovation environment for the development of digital financial inclusion. Additionally, traditional financial institutions should use emerging information technology to improve the drawbacks of traditional financial services and enhance their service efficiency, thus promoting innovation in the finance industry. The continuous innovation of digital financial inclusion can drive the innovation and development of the entire society.

Second, it is necessary to accelerate the construction of digital financial inclusion infrastructure, strengthen inter-regional exchanges and cooperation, and form a coordinated development pattern among regions. Strengthening infrastructure is conducive to giving full play to the role of digital financial inclusion in promoting economic growth in remote areas. This will further enhance the entrepreneurial vitality of underdeveloped areas and alleviate the imbalance in economic and social development. Strengthening the exchange and cooperation of digital financial inclusion between regions can improve the efficiency of resource allocation between regions. This could exert the spatial spillover effect of digital financial inclusion on the quality of economic growth. Therefore, relevant departments must ensure high-quality Internet connectivity, promote technologies such as big data and artificial intelligence, and support digital financial inclusion to accelerate economic development.

Third, it is essential to develop a regulatory system to prevent the possible risks associated with digital financial inclusion. Owing to the low service threshold and high openness of digital financial inclusion, the market is prone to information risk, credit risk, etc. Governmental departments must optimize laws and regulations related to digital financial inclusion to regulate market behavior and protect consumers' rights and interests. Additionally, the finance industry must improve its self-supervision and keep the bottom line of morality and risk, so as to promote the standardization, stabilization, and security development of digital financial inclusion. The improvement of the regulatory system is fundamental to the stable development of digital financial inclusion. Additionally, it ensures that digital financial inclusion promotes the quality of economic growth.

6.3. Future research directions

Presently, digital financial inclusion and technological innovation are undergoing rapid development. Therefore, future research can be conducted in the following directions. First, data of banks, securities, insurance, financial technology companies, etc., should be

collected—as much as possible. Thereafter, an index system should be constructed to comprehensively summarize the development level of digital financial inclusion. It would provide a more scientific and reasonable measurement index for the quantitative research of digital financial inclusion. Second, future research should investigate the risks associated with digital financial inclusion; construction of digital financial inclusion supervision system; improvement of laws and regulations related to digital financial inclusion; regulation of market behavior; and protection of consumers' rights. Third, the research on digital financial inclusion should delve into the micro level and explore the impacts of digital financial inclusion on enterprises and individuals, as well as its internal mechanism. Finally, digital financial inclusion includes various services, such as mobile payment, online loan, online insurance, online fund, etc. Therefore, future research on digital financial inclusion should inclusion should include different financial products and services.

Author contribution statement

Wenzhi Xi: Conceived and designed the experiments; Wrote the paper.

Yingdong Wang: Performed the experiments; Analyzed and interpreted the data; Contributed reagents, materials, analysis tools or data; Wrote the paper.

Data availability statement

Data will be made available on request.

Funding statement

We would like to thank editors and reviewers for their valuable suggestions. This study was supported by "the Fundamental Research Funds for the Central Universities", Zhongnan University of Economics and Law [Grant number: 202210563].

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Acknowledgements

We would like to thank the editors and reviewers for their valuable suggestions. This study was supported by "the Fundamental Research Funds for the Central Universities", Zhongnan University of Economics and Law [Grant number: 202210563].

References

- Z. Chen, Y. Li, Y. Wu, J. Luo, The transition from traditional banking to mobile internet finance: an organizational innovation perspective a comparative study of Citibank and ICBC, Financ. Innov. 3 (1) (2017) 12, https://doi.org/10.1186/s40854-017-0062-0.
- H. Bollaert, F. Lopez-de-Silanes, A. Schwienbacher, Fintech and access to finance, J. Corp. Finance 68 (2021), 101941, https://doi.org/10.1016/j. jcorpfin.2021.101941.
- [3] T.X. Sheng, The effect of fintech on banks' credit provision to SMEs: evidence from China, Finance Res. Lett. 39 (2021), 101558, https://doi.org/10.1016/j. frl.2020.101558.
- [4] P.K. Ozili, Impact of digital finance on financial inclusion and stability, Borsa Istanb. Rev. 18 (4) (2018) 329–340, https://doi.org/10.1016/j.bir.2017.12.003.
- [5] S.R. Das, The future of fintech, Financ. Manag. 48 (4) (2019) 981–1007, https://doi.org/10.1111/fima.12297.
- [6] I. Goldstein, W. Jiang, G.A. Karolyi, To FinTech and beyond, Rev. Financ. Stud. 32 (5) (2019) 1647–1661, https://doi.org/10.1093/rfs/hbz025.
 [7] A. Boot, P. Hoffmann, L. Laeven, L. Ratnovski, Fintech: what's old, what's new? J. Financ. Stabil. 53 (2021), 100836 https://doi.org/10.1016/j.
- jfs.2020.100836. [8] M.M. Hasan, Y.J. Lu, S. Khan, Promoting China's inclusive finance through digital financial services, Global Bus. Rev. 23 (4) (2022) 984–1006, https://doi.org/ 10.1177/0972150919895348.
- [9] P. Tantri, Fintech for the poor: financial intermediation without discrimination, Rev. Financ. 25 (2) (2021) 561–593, https://doi.org/10.1093/rof/rfaa039.
- [10] Q. Wang, J.B. Yang, Y.H. Chiu, T.Y. Lin, The impact of digital finance on financial efficiency, Manag. Decis. Econ. 41 (7) (2020) 1225–1236, https://doi.org/ 10.1002/mde.3168.
- [11] W. Li, W. Pang, Digital inclusive finance, financial mismatch and the innovation capacity of small and medium-sized enterprises: evidence from Chinese listed companies, Heliyon 9 (2) (2023), e13792, https://doi.org/10.1016/j.heliyon.2023.e13792.
- [12] J. Li, Y. Wu, J.J. Xiao, The impact of digital finance on household consumption: evidence from China, Econ, Model 86 (2020) 317–326, https://doi.org/ 10.1016/j.econmod.2019.09.027.
- [13] C. Yu, N. Jia, W. Li, R. Wu, Digital inclusive finance and rural consumption structure evidence from Peking University digital inclusive financial index and China household finance survey, China Agric, Econ. Rev. 14 (1) (2021) 165–183, https://doi.org/10.1108/caer-10-2020-0255.
- [14] L.-Y. Tay, H.-T. Tai, G.-S. Tan, Digital financial inclusion: a gateway to sustainable development, Heliyon 8 (6) (2022), e09766, https://doi.org/10.1016/j. heliyon.2022.e09766.
- [15] Q.X. Kong, D. Peng, Y.H. Ni, X.Y. Jiang, Z.Q. Wang, Trade openness and economic growth quality of China: empirical analysis using ARDL model, Finance Res. Lett. 38 (2021), 101488, https://doi.org/10.1016/j.frl.2020.101488.
- [16] D. Peng, R.R. Li, C.R. Shen, Z.Y. Wong, Industrial agglomeration, urban characteristics, and economic growth quality: the case of knowledge-intensive business services, Int. Rev. Econ. Finance 81 (2022) 18–28, https://doi.org/10.1016/j.iref.2022.05.001.
- [17] T.H. Li, X. Li, Does structural deceleration happen in China? Evidence from the effect of industrial structure on economic growth quality, Natl. Account. Rev. 2 (2) (2020) 155–173, https://doi.org/10.3934/nar.2020.2020009.
- [18] Y. Yang, S.H. Shi, J.J. Wu, Digital financial inclusion to corporation value: the mediating effect of ambidextrous innovation, Sustainability 14 (24) (2022), 16621, https://doi.org/10.3390/su142416621.

- [19] P. Khera, S. Ng, S. Ogawa, R. Sahay, Measuring digital financial inclusion in emerging market and developing economies: a new index, Asian Econ. Pol. Rev. 17 (2) (2022) 213–230, https://doi.org/10.1111/aepr.12377.
- [20] B.Y. Ren, L.Y. Li, H.M. Zhao, Y.B. Zhou, The financial exclusion in the development of digital finance a study based on survey data in the Jingjinji rural area, Singapore Econ. Rev. 63 (1) (2018) 65–82, https://doi.org/10.1142/s0217590818500017.
- [21] J. He, Q.H. Li, Can online social interaction improve the digital finance participation of rural households? China Agric. Econ. Rev. 12 (2) (2020) 295–313, https://doi.org/10.1108/caer-11-2019-0213.
- [22] G. Buchak, G. Matvos, T. Piskorski, A. Seru, Fintech, regulatory arbitrage, and the rise of shadow banks, J. Financ. Econ. 130 (3) (2018) 453–483, https://doi. org/10.1016/i.jfineco.2018.03.011.
- [23] Y.R. Li, G.L. Sun, Q. Gao, C.M. Cheng, Digital financial inclusion, financial efficiency and green innovation, Sustainability 15 (3) (2023) 1879, https://doi.org/ 10.3390/su15031879
- [24] P. Yue, A.G. Korkmaz, Z. Yin, H. Zhou, The rise of digital finance: financial inclusion or debt trap? Finance Res. Lett. 47 (2022) https://doi.org/10.1016/j. ftl.2021.102604.
- [25] Y.H. Yao, J.P. Li, X.L. Sun, Measuring the risk of Chinese Fintech industry: evidence from the stock index, Finance Res. Lett. 39 (2021), 101564, https://doi.org/ 10.1016/j.frl.2020.101564.
- [26] M. Cheng, Y. Qu, Does bank FinTech reduce credit risk? Evidence from China, Pac. Basin Finance J. 63 (2020), 101398, https://doi.org/10.1016/j. pacfin.2020.101398.
- [27] A. Zhang, S. Wang, B. Liu, P. Liu, How fintech impacts pre- and post-loan risk in Chinese commercial banks, Int. J. Financ. Econ. 27 (2) (2022) 2514–2529, https://doi.org/10.1002/ijfe.2284.
- [28] C. Li, S. He, Y. Tian, S. Sun, L. Ning, Does the bank's FinTech innovation reduce its risk-taking? Evidence from China's banking industry, J. Innov. Knowl. 7 (3) (2022), 100219, https://doi.org/10.1016/j.jik.2022.100219.
- [29] W.N.W. Azman-Saini, S.H. Law, A.H. Ahmad, FDI and economic growth: new evidence on the role of financial markets, Econ. Lett. 107 (2) (2010) 211–213, https://doi.org/10.1016/j.econlet.2010.01.027.
- [30] S. Asimakopoulos, Y. Karavias, The impact of government size on economic growth: a threshold analysis, Econ. Lett. 139 (2016) 65–68, https://doi.org/ 10.1016/j.econlet.2015.12.010.
- [31] A. Bick, Threshold effects of inflation on economic growth in developing countries, Econ. Lett. 108 (2) (2010) 126–129, https://doi.org/10.1016/j. econlet.2010.04.040.
- [32] T.O. Awokuse, Causality between exports, imports, and economic growth: evidence from transition economies, Econ. Lett. 94 (3) (2007) 389–395, https://doi. org/10.1016/j.econlet.2006.08.025.
- [33] Q. He, M.D. Bertness, J.F. Bruno, B. Li, G. Chen, T.C. Coverdale, A.H. Altieri, J. Bai, T. Sun, S.C. Pennings, J. Liu, P.R. Ehrlich, B. Cui, Economic development and coastal ecosystem change in China, Sci. Rep. 4 (2014) 5995, https://doi.org/10.1038/srep05995.
- [34] S. Bansal, G.D. Sharma, M.M. Rahman, A. Yadav, I. Garg, Nexus between environmental, social and economic development in South Asia: evidence from econometric models, Heliyon 7 (1) (2021), e05965, https://doi.org/10.1016/j.heliyon.2021.e05965.
- [35] M.T.I. Khan, S. Anwar, S.A. Sarkodie, M.R. Yaseen, A.M. Nadeem, Do natural disasters affect economic growth? The role of human capital, foreign direct investment, and infrastructure dynamics, Heliyon 9 (1) (2023), e12911, https://doi.org/10.1016/j.heliyon.2023.e12911.
- [36] Y. Yang, X. Lu, J. Chen, N. Li, Factor mobility, transportation network and green economic growth of the urban agglomeration, Sci. Rep. 12 (1) (2022), 20094, https://doi.org/10.1038/s41598-022-24624-5.
- [37] P. Zhang, Happiness inequality has a Kuznets-style relation with economic growth in China, Sci. Rep. 12 (1) (2022), 15712, https://doi.org/10.1038/s41598-022-19881-3.
- [38] X.H. Huang, B.Q. Cai, Y.L. Li, Evaluation index system and measurement of high-quality development in China, Rev. Cercet. Interv. Soc. 68 (2020) 163–178, https://doi.org/10.33788/rcis.68.11.
- [39] Q. Kong, D. Peng, Y. Ni, X. Jiang, Z. Wang, Trade openness and economic growth quality of China: empirical analysis using ARDL model, Finance Res. Lett. 101488 (2021), https://doi.org/10.1016/j.frl.2020.101488.
- [40] H. Zheng, Y. He, How does industrial co-agglomeration affect high-quality economic development? Evidence from Chengdu-Chongqing Economic Circle in China, J. Clean. Prod. 371 (2022), 133485, https://doi.org/10.1016/j.jclepro.2022.133485.
- [41] Q.X. Kong, R.R. Li, X. Jiang, P.B. Sun, D. Peng, Has transportation infrastructure development improved the quality of economic growth in China?s cities? A quasi-natural experiment based on the introduction of high-speed rail, Res. Int. Bus. Finance 62 (2022), 101726, https://doi.org/10.1016/j.ribaf.2022.101726.
- [42] Y.Y. Li, W.J. Liu, Spatial effects of environmental regulation on high-quality economic development: from the perspective of industrial upgrading, Front. Public Health 11 (2023), 1099887, https://doi.org/10.3389/fpubh.2023.1099887.
- [43] T. Lin, L.J. Wang, J.B. Wu, Environmental regulations, green technology innovation, and high-quality economic development in China: application of mediation and threshold effects, Sustainability 14 (11) (2022) 6882, https://doi.org/10.3390/su14116882.
- [44] J. Wang, G.X. Zhang, Can environmental regulation improve high-quality economic development in China? The mediating effects of digital economy,
- Sustainability 14 (19) (2022), 12143, https://doi.org/10.3390/su141912143.
 [45] M. Chen, H.B. Wang, Import technology sophistication and high-quality economic development: evidence from city-level data of China, Econ Res-Ekon Istraz 35 (1) (2022) 1106–1141, https://doi.org/10.1080/1331677x.2021.1956362.
- [46] L. Agnello, S.K. Mallick, R.M. Sousa, Financial reforms and income inequality, Econ. Lett. 116 (3) (2012) 583-587, https://doi.org/10.1016/j.
- econlet.2012.06.005. [47] D. Asteriou, K. Spanos, The relationship between financial development and economic growth during the recent crisis: evidence from the EU, Finance Res. Lett.
- 28 (2019) 238–245, https://doi.org/10.1016/j.frl.2018.05.011. [48] Z. Wong, R.R. Li, Y.D. Zhang, Q.X. Kong, M. Cai, Financial services, spatial agglomeration, and the quality of urban economic growth-based on an empirical
- analysis of 268 cities in China, Finance Res. Lett. 43 (2021), 101993, https://doi.org/10.1016/j.frl.2021.101993. [49] J. Gao, D.L. Wu, Q. Xiao, A. Randhawa, Q. Liu, T. Zhang, Green finance, environmental pollution and high-quality economic development - a study based on
- China's provincial panel data, Environ. Sci. Pollut. Res. (2022), https://doi.org/10.1007/s11356-022-24428-0. [50] X.H. Chen, Y.Q. He, The impact of financial resilience and steady growth on high-quality economic development - based on a heterogeneous intermediary effect
- analysis, Sustainability 14 (22) (2022), 14748, https://doi.org/10.3390/su142214748.
- [51] Y. Liu, L. Luan, W.L. Wu, Z.Q. Zhang, Y. Hsu, Can digital financial inclusion promote China's economic growth? Int. Rev. Financ. Anal. 78 (2021), 101889 https://doi.org/10.1016/j.irfa.2021.101889.
- [52] N. Yu, Y. Wang, Can digital inclusive finance narrow the Chinese urban-rural income gap? The perspective of the regional urban-rural income structure, Sustainability 13 (11) (2021) 6427, https://doi.org/10.3390/su13116427.
- [53] F. Xu, X.R. Zhang, D. Zhou, Does digital financial inclusion reduce the risk of returning to poverty? Evidence from China, Int. J. Financ. Econ. (2023), https:// doi.org/10.1002/iife.2812.
- [54] V. Murinde, Financial development and economic growth: global and African evidence, J. Afr. Econ. 21 (2012) II0–I56, https://doi.org/10.1093/jae/ejr042.
- [55] F.D. Duarte, A.P.M. Gama, M.A. Gulamhussen, Credit risk, owner liability, and bank loan maturities during the global financial crisis, Eur. Financ. Manag. 26 (3) (2020) 628–683, https://doi.org/10.1111/eufm.12239.
- [56] G. Yeung, How banks in China make lending decisions, J. Contemp. China 18 (59) (2009) 285-302, https://doi.org/10.1080/10670560802576034.
- [57] A. Khafagy, Finance, distribution and the economic objective of financial cooperative institutions, Ann. Public Coop. Econ. 90 (3) (2019) 487–511, https://doi. org/10.1111/apce.12216.
- [58] L.A. Beisland, K.O. Djan, R. Mersland, T. Randoy, Measuring social performance in social enterprises: a global study of microfinance institutions, J. Bus. Ethics 171 (1) (2021) 51–71, https://doi.org/10.1007/s10551-019-04417-z.

- [59] E. Scannella, What drives the disintegration of the loan origination value chain in the banking business, Bus. Process Manag. J. 21 (2) (2015) 288–311, https:// doi.org/10.1108/bpmj-02-2014-0017.
- [60] F. Sardo, Z. Serrasqueiro, M.R. Armada, The importance of owner loans for rebalancing the capital structure of small knowledge-intensive service firms, Res. Int. Bus. Finance 61 (2022), 101657, https://doi.org/10.1016/j.ribaf.2022.101657.
- [61] Y. Zhao, D. Li, L.Q. Pan, Cooperation or competition: an evolutionary game study between commercial banks and big data-based e-commerce financial institutions in China, Discrete Dynam Nat. Soc. (2015), 890972, https://doi.org/10.1155/2015/890972, 2015.
- [62] Q.L. Zhang, D.H. Yang, J.Y. Qin, Multi-Party evolutionary game analysis of accounts receivable financing under the application of central bank digital currency, J. Theor. Appl. Electron. Commer. Res. 18 (1) (2023) 394–415, https://doi.org/10.3390/jtaer18010021.
- [63] A. Pelletier, S. Khavul, S. Estrin, Innovations in emerging markets: the case of mobile money, Ind. Corp. Change 29 (2) (2020) 395–421, https://doi.org/ 10.1093/icc/dtz049
- [64] W. He, J.L. Hung, L.X. Liu, Impact of big data analytics on banking: a case study, J. Enterprise Inf. Manag. 36 (2) (2023) 459–479, https://doi.org/10.1108/ ieim-05-2020-0176.
- [65] J. Xue, D. Zhu, L.J. Zhao, C.C. Wang, H.Y. Li, Redundancy identification and optimization scheme of branches for sustainable operation of commercial banks, Sustainability 11 (15) (2019) 4111, https://doi.org/10.3390/su11154111.
- [66] H.-B. Ong, S. Wasiuzzaman, L.-L. Chong, S.-W. Choon, Digitalisation and financial inclusion of lower middle-income ASEAN, Heliyon 9 (2) (2023), e13347, https://doi.org/10.1016/j.heliyon.2023.e13347.
- [67] X.L. Yang, Y.D. Huang, M. Gao, Can digital financial inclusion promote female entrepreneurship? Evidence and mechanisms, N. Am. J. Econ. Finance 63 (2022), https://doi.org/10.1016/j.najef.2022.101800.
- [68] W.W. Xie, T. Wang, X. Zhao, Does digital inclusive finance promote coastal rural entrepreneurship? J. Coast Res. (2020) 240–245, https://doi.org/10.2112/ si103-052.1.
- [69] T. Gries, W. Naude, Entrepreneurship, structural change and a global economic crisis, Enterpren. Res. J. 1 (3) (2011), https://doi.org/10.2202/2157-5665.1029.
- [70] F. Noseleit, Entrepreneurship, structural change, and economic growth, J. Evolutionary Econ. 23 (4) (2013) 735–766, https://doi.org/10.1007/s00191-012-0291-3.
- [71] E.L. Glaeser, S.P. Kerr, W.R. Kerr, Entrepreneurship and urban growth: an empirical assessment with historical mines, Rev. Econ. Stat. 97 (2) (2015) 498–520, https://doi.org/10.1162/REST a 00456.
- [72] T. Beck, H. Pamuk, R. Ramrattan, B.R. Uras, Payment instruments, finance and development, J. Dev. Econ. 133 (2018) 162–186, https://doi.org/10.1016/j. jdeveco.2018.01.005.
- [73] Y. Shen, C.J. Hueng, W.X. Hu, Measurement and spillover effect of digital financial inclusion: a cross-country analysis, Appl. Econ. Lett. 28 (20) (2021) 1738–1743, https://doi.org/10.1080/13504851.2020.1853663.
- [74] G.J. Zhang, Y. Chen, G.N. Wang, C.S. Zhou, Spatial-temporal evolution and influencing factors of digital financial inclusion: county-level evidence from China, Chin. Geogr. Sci. (2023), https://doi.org/10.1007/s11769-023-1333-5.
- [75] X. Wang, X. Wang, X.H. Ren, F.H. Wen, Can digital financial inclusion affect CO2 emissions of China at the prefecture level? Evidence from a spatial econometric approach, Energy Econ. 109 (2022), 105966, https://doi.org/10.1016/j.eneco.2022.105966.
- [76] B.Q. Lin, Y.C. Zhou, Does energy efficiency make sense in China? Based on the perspective of economic growth quality, Sci. Total Environ. 804 (2022), 149895, https://doi.org/10.1016/j.scitotenv.2021.149895.
- [77] X. Gong, Y. Wang, B.Q. Lin, Assessing dynamic China's energy security: based on functional data analysis, Energy 217 (2021), https://doi.org/10.1016/j. energy.2020.119324.
- [78] B.E. Hansen, Threshold effects in non-dynamic panels: estimation, testing, and inference, J. Econom. 93 (2) (1999) 345–368, https://doi.org/10.1016/S0304-4076(99)00025-1.
- [79] M.H. Seo, S. Kim, Y.-J. Kim, Estimation of dynamic panel threshold model using Stata, STATA J. 19 (3) (2019) 685–697, https://doi.org/10.1177/ 1536867x19874243.
- [80] M.H. Seo, Y. Shin, Dynamic panels with threshold effect and endogeneity, J. Econom. 195 (2) (2016) 169–186, https://doi.org/10.1016/j. jeconom.2016.03.005.
- [81] G. Li, R. Zhang, S. Feng, Y. Wang, Digital finance and sustainable development: evidence from environmental inequality in China, Bus. Strat. Environ. 31 (7) (2022) 3574–3594, https://doi.org/10.1002/bse.3105.
- [82] S. Ren, L. Li, Y. Han, Y. Hao, H. Wu, The emerging driving force of inclusive green growth: does digital economy agglomeration work? Bus. Strat. Environ. 31 (4) (2022) 1656–1678, https://doi.org/10.1002/bse.2975.
- [83] Q. Xu, M. Zhong, M. Cao, Does digital investment affect carbon efficiency? Spatial effect and mechanism discussion, Sci. Total Environ. 827 (2022), 154321, https://doi.org/10.1016/j.scitotenv.2022.154321.
- [84] C.C. Lee, R.C. Lou, F.H. Wang, Digital financial inclusion and poverty alleviation: evidence from the sustainable development of China, Econ, Anal. Policy 77 (2023) 418–434, https://doi.org/10.1016/j.eap.2022.12.004.
- [85] I. Ozturk, S. Ullah, Does digital financial inclusion matter for economic growth and environmental sustainability in OBRI economies? An empirical analysis, Resour. Conserv. Recycl. 185 (2022), 106489, https://doi.org/10.1016/j.resconrec.2022.106489.
- [86] Y. Chen, S.P. Yang, Q. Li, How does the development of digital financial inclusion affect the total factor productivity of listed companies? Evidence from China, Finance Res. Lett. 47 (2022) 6, https://doi.org/10.1016/j.frl.2022.102956.
- [87] Y. Liu, M. Liu, G.G. Wang, L.L. Zhao, P. An, Effect of environmental regulation on high-quality economic development in China an empirical analysis based on dynamic spatial Durbin model, Environ. Sci. Pollut. Res. 28 (39) (2021) 54661–54678, https://doi.org/10.1007/s11356-021-13780-2.
- [88] Y.L. Li, M.X. Wang, G.K. Liao, J.X. Wang, Spatial spillover effect and threshold effect of digital financial inclusion on farmers' income growth-based on provincial data of China, Sustainability 14 (3) (2022) 1838, https://doi.org/10.3390/su14031838.
- [89] A. Aziz, U. Naima, Rethinking digital financial inclusion: evidence from Bangladesh, Technol. Soc. 64 (2021), 101509, https://doi.org/10.1016/j. techsoc.2020.101509.